Author's response to reviews

Title: Evaluation of effectiveness of class-based nutrition intervention on changes in soft drink and milk consumption among young adults.

Authors:

Eun-Jeong Ha (eha@kent.edu)
Natalie Caine-Bish (ncaine@kent.edu)
Christopher Holloman (holloman@stat.osu.edu)
Karen Lowry-Gordon (klowry@kent.edu)

Version: 5 Date: 22 October 2009

Author's response to reviews:

To Whom It May Concern:

Thank you for the valuable suggestions for revision, which made our manuscript more suitable for a publication. As you notice, we have revised the manuscript based on the comments made by your editorial production team. Attached is our point-by-point response to the suggestions.

If there is anything you need further, please do not hesitate to contact me at eha@kent.edu or 330-672-2701.

Thank you very much.

Sincerely yours,

Eun-Jeong Ha, Ph.D.
Assistant Professor
Nutrition and Dietetics

Major revisions (We require the author to make these changes):

Author list - The list of authors in the manuscript should be written exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system.

Author list has been checked and proper change has been made.

Authors' contributions - Please indicate that all authors read and approved the final manuscript.

The statement indicating that all authors read and approved the final manuscript
has been added.

Table vertical elements - Please note that we are unable to display vertical rules within the tables - please re-layout your table without these vertical elements. We are unable to correctly display merged cells where the merged cell crosses rows.

Vertical lines in the table were removed.

References - all references must be cited in consecutive numerical order to ensure all citations link up correctly in the final publication. Currently we notice reference 17 is out of order. Please amend this.

References were all checked to and rearranged in order.

Please ensure all the references are cited in square brackets. Currently some are cited in round brackets.

All the references now are in square brackets.

Minor revisions (We can make these changes on resubmission, although it will speed up publication of your manuscript if you do them while making the major changes above):

Line numbering - please remove this.

Line numbers were removed.

Please remove the 'title' sub heading from the title page.

'title' sub heading was removed from the title page.

Additional file - Please delete the additional files that contain your response to the reviewers.

Additional files containing our response to the reviewers will be deleted when we submit this revision.

Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions as this is the final proofing stage for your manuscript, and you will not be able to make changes after acceptance.

We appreciate that we had one more opportunity to make some minor, editorial changes in our manuscript.